Author's response to reviews

Title: Mothers' AdvocateS In the Community (MOSAIC) - non-professional mentor support to reduce intimate partner violence and depression in mothers: a cluster randomised trial in primary care

Authors:

Angela J Taft (a.taft@latrobe.edu.au)
Rhonda Small (r.small@latrobe.edu.au)
Kelsey L Hegarty (k.hegarty@unimelb.edu.au)
Lyndsey F Watson (l.watson@latrobe.edu.au)
Lisa Gold (lisa.gold@deakin.edu.au)
Judith A Lumley (j.lumley@latrobe.edu.au)

Version: 3 Date: 15 February 2011

Author's response to reviews: see over
Dear editors

Re: MS: 5470133514468596
Mothers' AdvocateS In the Community (MOSAIC) - non-professional mentor support to reduce intimate partner violence and depression in mothers: a cluster randomised trial in primary care Angela J Taft, Rhonda Small, Kelsey L Hegarty, Lyndsey F Watson, Lisa Gold and Judith A Lumley

We are very pleased at the possibility that this outcomes paper will be available soon to other intervention triallists in this field and those interested in policy and programs on violence against women and children.

Corrections:

Re M Kiely revisions

P.4 The verb has been altered to ‘was’ and upskilling to ‘training’.

P.11 now reads PS rather than the full term ‘propensity score’.

Re Leslie Davidson revisions

P. 5 ‘women’ has been altered to ‘mentors’

P.6 the bullet has been added.

P.7 As indicated in the text, we are unable to reference this article as it has neither been written nor published.

P.9 now reads ‘had taken up new studies or training or returned to further training or education over the past 12 months compared with non-mentored women’

Editors’ revisions

The number 790 has been spelt out.

I have made a few tiny corrections (removing italics, spaces etc) which are also indicated.

We now trust that the manuscript is acceptable for publication in your journal.

Angela Taft (for the author group).